Author's response to reviews

Title: Emergency vaccination of rabies under limited resources - combating or containing?

Authors:

Dirk Eisinger (dirk.eisinger@ufz.de)
Hans-Hermann Thulke (hans.thulke@ufz.de)
Thomas Selhorst (thomas.selhorst@wus.bfav.de)
Thomas Mueller (thomas.mueller@wus.bfav.de)

Version: 5 Date: 3 March 2005

Author's response to reviews: see over
Changes within Resubmission 5 of:

Article title: Emergency vaccination of rabies under limited resources - combating or containing?
MS ID : 1161983425409182
Authors : Dirk Eisinger, Hans-Hermann Thulke, Thomas Müller and Thomas Selhorst
Journal : BMC Infectious Diseases

Comments by Graham C. Smith:
All seven minor revisions of the second review by Graham C Smith dating from the 18th February 2005 have been incorporated without changes.

Other changes:
A clarification was added in the header of Table 2.